Author's response to reviews

Title: Comparative analysis of oral treponeme bacteria associated with periodontal health and disease

Authors:

Meng You (fiona_scu@yahoo.com.cn)
Sisu Mo (mosisu@126.com)
W Keung Leung (ewkleung@hkucc.hku.hk)
Rory M. Watt (rmwatt@hku.hk)

Version: 2 Date: 8 January 2013

Author's response to reviews:

Dear Editor,

In the revised manuscript, two sentences have been added to the beginning of the Methods section. These clearly state that IRB ethical approval was obtained, all subjects gave informed consent, and the study conformed to the Helsinki guidelines. The rest of the manuscript remains entirely unchanged.

Yours sincerely

Rory Watt